Author’s response to reviews

Title: Otitis media in children vaccinated with either 7-valent or 10-valent pneumococcal conjugate vaccines: a cross sectional comparison

Authors:

Amanda J Leach (Amanda.Leach@menzies.edu.au)
Christine Wigger (Christine.Wigger@menzies.edu.au)
Ross Andrews (Ross.Andrews@menzies.edu.au)
Mark Chatfield (mark.chatfield@menzies.edu.au)
Heidi Smith-Vaughan (heidi.smith-vaughan@menzies.edu.au)
Peter S Morris (peter.morris@menzies.edu.au)

Version: 8 Date: 25 July 2014

Author’s response to reviews:

the following duplicated information has been removed from the main text (page 8)

Funding: The study was funded by the Australian National Health and Medical Research Council (project #545232) for the years 2008 to 2010, and by GlaxoSmithKline for the years 2009 to 2012.